Author's response to reviews

Title: Gene by Environment QTL mapping through Multiple Trait Analyses in blood pressure salt-sensitivity: identification of a novel QTL in rat chromosome 5

Authors:

Julia Pavan (pavan@ime.usp.br)
Alexandre C Pereira (lbmpereira@incor.usp.br)
Cesar H Torres (chtorres@linux.ime.usp.br)
Jose E Krieger (krieger@incor.usp.br)

Version: 7 Date: 16 May 2006

Author's response to reviews:

Dear Editor,

We have revised the article on the two points required by the editorial board:

1. Corrected the website references;
2. Corrected the competing interests paragraph.

Thank you very much for your assistance during the editorial process of this manuscript.

Sincerely,

Alexandre Pereira